Author's response to reviews

Title: Petroleum Contaminated Water and Health Outcomes in a Rural Nigerian Community: A Cross-Sectional Pilot Study

Authors:

Kalé Z Kponee (kak785@mail.harvard.edu)
Andrea Chiger (achiger@bu.edu)
Iyenemi I Kakulu (ibkakulu@hotmail.com)
Donna Vorhees (dvorhees@bu.edu)
Wendy Heiger-Bernays (whb@bu.edu)

Version: 2  
Date: 11 June 2015

Author's response to reviews:

As our research involves humans we include a statement of ethical approval in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number from the Boston University Medical Center, in compliance with the Helsinki Declaration.

We include a figure title and legend section after the reference list.